Author’s response to reviews

Title: The relationship between facial skeletal class and expert-rated interpersonal skill: an epidemiological survey on young Italian adults

Authors:
- Andrea Senna (andreasenna75@hotmail.com)
- Domenico Abbenante (dabbenante@hotmail.it)
- Lucio Tremolizzo (lucio.tremolizzo@unimib.it)
- Guglielmo Campus (gcampus@uniss.it)
- Laura Strohmenger (laura.strohmenger@unimi.it)

Version: 5 Date: 7 October 2006

Author’s response to reviews: see over
Dear Dr. Appleford,

Please find enclosed the formatted version of the manuscript “The relationship between facial skeletal class and expert-rated interpersonal skill: an epidemiological survey on young Italian adults” by Senna et al.

As required, a point-by-point description of the changes follows:

1. The title is now in sentence case  
2. The affiliations have been opportunely corrected  
3. Keywords and counts have been removed from the manuscript title page  
4. In the Abstract the subheading “Purpose” has been renamed as “Background”  
5. The “Introduction” has been renamed as “Background”  
6. “Materials & Methods” has been renamed as “Methods”  
7. A “Conclusions” section has been inserted after the “Discussion”  
8. A title has been placed above each table  
9. The image files do not include the title or figure number  
10. The figures have been cropped as closely as possible to minimize white space around the image.

Finally, I would like to state that all the authors of the byline approved the manuscript content and that this manuscript is both, not under consideration for publication elsewhere, and free of any conflict of interest or financial implications.

Sincerely,

Guglielmo Campus, DDS, PhD  
Assistant Clinical Professor  
Dental Institute - University of Sassari - Italy